Author’s response to reviews

Title: Streptococcus agalactiae infective endocarditis complicated by multiple mycotic hepatic aneurysms and massive splenic infarction: a case report

Authors:

Pietro Achilli (pietro.achilli89@gmail.com)
Angelo Guttadauro (angelo.guttadauro@tiscali.it)
Paolo Bonfanti (pa.bonfanti@asst-lecco.it)
Sabina Terragni (sabina.terragni@gmail.com)
Luca Fumagalli (lu.fumagalli@asst-lecco.it)
Ugo Cioffi (ugo.cioffi@guest.unimi.it)
Francesco Gabrielli (francesco.gabrielli@unimib.it)
Matilde De Simone (matilde.desimone@alice.it)
Marco Chiarelli (marco.chiarelli@gmail.com)

Version: 2 Date: 12 Dec 2017

Author’s response to reviews:

BMGE-D-17-00219R1

Streptococcus agalactiae infective endocarditis complicated by multiple mycotic hepatic aneurysms and massive splenic infarction: a case report

Pietro Achilli, MD; Angelo Guttadauro; Paolo Bonfanti, MD; Sabina Terragni, MD; Luca Fumagalli, MD; Ugo Cioffi, MD, PhD; Francesco Gabrielli, MD; Matilde De Simone, MD, PhD; Marco Chiarelli, MD

BMC Gastroenterology

Dear Dr. Guttadauro,

Your manuscript "Streptococcus agalactiae infective endocarditis complicated by multiple mycotic hepatic aneurysms and massive splenic infarction: a case report" (BMGE-D-17-00219R1) has been assessed by our reviewers. Based on these reports, and my own assessment
as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Gastroenterology, once you have carried out some essential revisions below:

1. We noted that the corresponding author in the manuscript and in the submission system is different. Please ensure to provide the correct corresponding author.

2. We noted that in your revised manuscript, there are not uploaded figures. Please upload the figures and ensure they are cited.

3. Please remove the point to point response from the file inventory because it is no longer necessary at this stage.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Once you have made the necessary corrections, please submit a revised manuscript online at:

http://bmge.edmgr.com/